Author's response to reviews

Title: Interventions to improve benzodiazepine prescribing, lessons from the past 20 years to guide future interventions

Authors:

Alesha J Smith (alesha.smith@uq.edu.au)
Susan E Tett (s.tett@uq.edu.au)

Version: 6 Date: 30 November 2010

Author's response to reviews:

30 November 2010

Dear BMC Series Editorial Production Team

Thank you for the format changes. The responses to your requests are in red below.

Title: please ensure that the title is identical in the manuscript and the submission system – This has been changed in the submission system.

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings:
Background, Methods, Results, Conclusions Please remember to also update the abstract details on the submission page. Both the manuscript and abstract in the submission system have been updates to comply with this format.

Text boxes: unfortunately text boxes are not supported, please include this information as plain text, you can use indentation to highlight it – The text box has been removed.

References: References should be cited in the text using consecutive numbers in square brackets. Currently we can not find references 57-70 cited – The references have been double checked, references 57-70 are in the tables.

Tables: we notice that you have included the tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references. Tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. – The tables have been included in the manuscript after the references and formatted to ‘simple 1’.

Introduction: please rename this section Background - This has been changed.
If you need further changes or have any further questions, please do not hesitate to contact me.

We look forward to seeing the article proofs

Sincerely,

Alesha Smith